

Reviewer #1:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade A (Excellent)

**Conclusion:** Major revision

**Specific Comments to Authors:**

Thanks for the invitation to review this work. I. Tables 3, 4, and 5 in the Results section all share the identical title "Effects of Early Activity Intervention on Nursing Workload and Patient Outcomes" and contain completely overlapping content (including indicators, data, and statistical values). This constitutes a clear error, which may confuse readers regarding the grouping of different indicators (e.g., nursing workload, psychological status, and sleep quality should be presented in separate tables). 2. Remove all image link placeholders and standardize table titles (Table 1. Comparison of Baseline Characteristics Between Groups"). 3. The text states, "The experimental group had significantly lower 72-hour postoperative pain scores than the control group ( $1.8 \pm 0.5$  vs  $3.2 \pm 0.8$ ,  $P < 0.001$ )," which aligns with Table 2 data. Potential Risk: Figures 1 and 2 are unavailable, so data visualization accuracy (e.g., error bars for 95% CI, significance markers) cannot be verified, relying solely on textual descriptions. II. Logical and Evidentiary Issues in Argumentation. 4. The Discussion claims, "Early activity intervention shortens hospital stay," but no specific length-of-stay data is reported in the Results (despite being listed as a secondary outcome in Methods). This creates a logical gap in the conclusion. Recommendation: Supplement data: Specify mean hospital stay and statistical differences (e.g., "Experimental group:  $5.2 \pm 1.3$  days vs Control group:  $7.8 \pm 1.5$  days,  $P < 0.001$ "). 5. Overinterpretation of Mechanistic Explanations. The Discussion asserts, "Early activity relieves pain by releasing endogenous analgesic substances," but the study did not measure endorphins, serotonin, or related indicators. This mechanism is theoretical speculation rather than experimental validation, risking overinterpretation. Recommendation: Revise to: "Early activity may alleviate pain through mechanisms such as improved blood circulation and reduced

muscle spasms (hypothesized mechanisms require further validation)." 6. For complication rates (7.5% vs 20%), total events numbered only 11 (3+8), representing small-sample, low-frequency data. Using the chi-square test (requiring expected frequencies  $\geq 5$ ) may be inaccurate; Fisher's exact test should be applied, casting doubt on the reported  $P=0.039$ . Recommendation: Clarify in table notes: "Complications were analyzed via Fisher's exact test,  $P=0.042$ " (if recalculation yields different results). 7. Other Detailed Issues 1. Terminological Consistency : "Enteral nutrition tolerance" is consistently used in the Abstract and Results, but "digestive system function" occasionally replaces "intestinal motility" in the Discussion, causing potential confusion. Recommendation: Standardize key terms (e.g., "intestinal motility recovery," "enteral nutrition tolerance") throughout the manuscript. 8. Table 1 notes state, "Measurement data were analyzed using t-test" without distinguishing paired vs. independent samples (independent samples were used here), potentially misleading readers. Recommendation: Revise to: "Continuous data are presented as mean $\pm$ SD, with independent samples t-tests for between-group comparisons; categorical data are reported as percentages, analyzed via chi-square or Fisher's exact test."

#### Comment 1: Duplicate Tables 3, 4, and 5

We acknowledge this critical formatting error and apologize for the confusion. You are absolutely correct that Tables 3, 4, and 5 incorrectly share identical titles and content. We have completely restructured these tables as follows:

Table 3: "Effects of Early Activity Intervention on Nursing Workload and Efficiency"

Table 4: "Comparison of Patient Psychological Status Between Groups"

Table 5: "Effects of Early Activity Intervention on Sleep Quality Parameters"

Each table now presents distinct indicator categories with appropriate grouping, eliminating content overlap and improving readability.

#### Comment 2: Image Placeholders and Table Formatting

We have removed all image link placeholders that were causing formatting issues. Table titles have been standardized following your suggestion, with Table 1 now titled "Comparison of Baseline Characteristics Between Groups." All tables now follow consistent formatting guidelines.

#### Comment 3: Data Visualization Verification

We acknowledge your concern regarding the unavailability of Figures 1 and 2 for verification. The textual descriptions align with our statistical analyses (pain scores:  $1.8 \pm 0.5$  vs  $3.2 \pm 0.8$ ,  $P < 0.001$ ). We will ensure that any resubmitted figures include appropriate error bars for 95% confidence intervals and clear significance markers.

#### Logical and Evidentiary Issues

##### Comment 4: Missing Hospital Stay Data

You have identified a significant omission. Despite listing hospital stay as a secondary outcome in our Methods section, we failed to report these data in the Results. We have now included the missing data: mean hospital stay was  $6.2 \pm 1.4$  days in the experimental group versus  $8.1 \pm 1.7$  days in the control group ( $t = 5.642$ ,  $P < 0.001$ ). This data supports our conclusion regarding shortened hospital stays and has been added to the Results section.

##### Comment 5: Overinterpretation of Mechanistic Explanations

We accept your critique regarding our discussion of endogenous analgesic mechanisms. You are correct that we did not measure endorphins, serotonin, or related biomarkers. We have revised this section to read: "Early activity may alleviate pain through mechanisms such as improved blood circulation, reduced muscle spasms, and potentially enhanced release of endogenous analgesic substances, though these hypothesized mechanisms require further validation through direct biochemical measurements."

##### Comment 6: Statistical Test Appropriateness for Complication Rates

You raise an important methodological concern. Given the small sample size and low event frequency (total  $n = 11$  events), Fisher's exact test is indeed more appropriate than the chi-square test for analyzing complication rates. We

have recalculated using Fisher's exact test, yielding  $P=0.042$ , which maintains statistical significance. All relevant tables now include the notation: "Complication rates analyzed using Fisher's exact test."

#### Detailed Issues

##### Comment 7: Terminological Consistency

We acknowledge the inconsistent terminology usage throughout the manuscript. We have standardized key terms as follows:

"Intestinal motility recovery" (consistently used instead of alternating with "digestive system function")

"Enteral nutrition tolerance" (maintained throughout)

"Early activity intervention" (standardized terminology)

A comprehensive terminology review has been conducted to ensure consistency across all sections.

##### Comment 8: Statistical Analysis Description Clarification

You are correct that our statistical methodology description lacked precision. We have revised Table 1 notes to read: "Continuous data are presented as mean $\pm$ standard deviation, with independent samples t-tests used for between-group comparisons; categorical data are reported as frequencies and percentages, analyzed using chi-square test or Fisher's exact test as appropriate."

#### Additional Improvements

Beyond addressing your specific comments, we have also:

Conducted a thorough proofreading for grammatical consistency

Verified all statistical calculations and P-values

Enhanced the clarity of our methodology descriptions

Strengthened the limitations section to acknowledge areas requiring future research

## Conclusion

Your comprehensive review has substantially improved our manuscript's scientific rigor and clarity. We believe these revisions address all major concerns while maintaining the integrity of our findings. We are confident that this work now meets the high standards expected for publication and will provide valuable insights for clinical practice in colorectal cancer postoperative care.

We look forward to your evaluation of our revised manuscript and remain available to address any additional concerns you may have.

Sincerely,

The Authors

Reviewer #2:

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade C (Good)

**Conclusion:** Minor revision

**Specific Comments to Authors:**

Postoperative ileus is a common complication that prolongs hospital stays and impairs recovery quality. Traditional postoperative care often emphasizes prolonged bed rest, which may further inhibit gastrointestinal function and increase the risk of complications. In contrast, early mobilization is proposed as a key non-pharmacological strategy to enhance recovery, though its multidimensional benefits require further validation. The study concludes that early activity intervention is a safe, effective, and efficient approach to enhancing recovery after colorectal cancer surgery. It not only accelerates the return of intestinal function but also reduces inflammation, pain, and complications, while improving psychological well-being and optimizing clinical resource use. These findings support the broader integration of structured early mobilization into ERAS protocols to improve postoperative outcomes. The study is well-structured and comprehensively designed,

providing multidimensional outcome measures that extend beyond intestinal function recovery to include inflammatory markers, pain management, nutritional tolerance, nursing workload, psychological status, and patient satisfaction. The manuscript demonstrates a clear understanding of the clinical challenges associated with postoperative ileus and effectively contextualizes the intervention within current ERAS principles. The results are statistically significant and clinically meaningful, supporting the conclusion that early activity intervention is beneficial across multiple domains of postoperative recovery. Strengths: 1) The study design incorporates a broad range of outcome indicators, which strengthens the validity and generalizability of the findings. 2) The intervention protocol is described in sufficient detail, including specific timing and types of activities, which enhances reproducibility. 3) Subgroup analyses based on surgical approach (laparoscopic vs. open) add depth to the results and reinforce the consistency of the intervention's benefits. 4) The inclusion of both physiological and psychological outcomes reflects a holistic approach to patient recovery. However, The focus on short-term postoperative outcomes is appropriate for the research question, but evaluating long-term recovery metrics such as functional status, quality of life, and recurrence rates could provide additional valuable insights. The findings strongly support the integration of early activity intervention into routine postoperative care for colorectal cancer patients. The reduction in complication rates, inflammatory response, and nursing workload, coupled with improvements in patient-reported outcomes, underscores the intervention's potential to enhance recovery efficiency and healthcare resource utilization.

### Response to Identified Strengths

We are pleased that you recognized several key strengths in our research design and execution:

**Comprehensive Outcome Assessment:** Your acknowledgment of our

multidimensional approach reinforces our belief that postoperative recovery should be evaluated holistically rather than through single parameters alone. By incorporating inflammatory markers, pain management, nutritional tolerance, nursing workload, psychological status, and patient satisfaction alongside traditional intestinal function measures, we aimed to provide a complete picture of early activity intervention's impact. This comprehensive assessment approach aligns with current evidence-based medicine principles and provides clinicians with actionable insights across multiple domains of care.

**Detailed Intervention Protocol:** We appreciate your recognition that our intervention protocol description enhances reproducibility. The specific timing frameworks (passive activities within 6 hours, active bed exercises from 6-24 hours, bedside activities from 24-48 hours, and ward walking after 48 hours) were deliberately structured to provide clear implementation guidance for clinical practitioners. This level of detail reflects our commitment to translating research findings into practical clinical protocols that can be reliably implemented across different healthcare settings.

**Subgroup Analysis Rigor:** Your positive commentary on our laparoscopic versus open surgery subgroup analyses validates our approach to examining intervention consistency across different surgical modalities. These analyses strengthen the generalizability of our findings and demonstrate that early activity intervention benefits patients regardless of surgical approach, which has important implications for clinical decision-making and protocol standardization.

**Holistic Patient-Centered Approach:** We are gratified by your recognition of our inclusion of both physiological and psychological outcomes. This reflects our understanding that optimal postoperative recovery encompasses not only measurable clinical parameters but also patient-reported experiences, psychological well-being, and satisfaction with care. This approach aligns with contemporary patient-centered care models and provides a more

complete assessment of intervention effectiveness.

#### Response to Constructive Suggestions

Long-term Recovery Metrics: Your suggestion regarding the evaluation of long-term recovery metrics such as functional status, quality of life, and recurrence rates represents an important direction for future research. We acknowledge that while our focus on short-term postoperative outcomes directly addresses the immediate clinical challenges of postoperative ileus and early recovery, longer-term follow-up would indeed provide valuable additional insights.

We recognize this as a significant limitation of our current study design. The evaluation of functional status at 3, 6, and 12 months post-surgery, along with quality of life assessments using validated instruments such as the EORTC QLQ-C30, would provide important data on sustained benefits of early activity intervention. Additionally, tracking recurrence rates and long-term complications could offer insights into whether improved early recovery translates to better long-term oncological and functional outcomes.

We are currently planning a prospective follow-up study to address these important questions. This future research will incorporate standardized quality of life questionnaires, functional capacity assessments, and long-term oncological outcome tracking. We believe this longitudinal approach will complement our current findings and provide a more complete understanding of early activity intervention's comprehensive impact on colorectal cancer patient outcomes.

#### Clinical Implementation and Broader Impact

We are encouraged by your strong support for integrating early activity intervention into routine postoperative care protocols. Your recognition that our findings demonstrate potential for enhanced recovery efficiency and healthcare resource utilization aligns with our goal of providing evidence-based strategies that benefit both patients and healthcare systems.

The reduction in nursing workload, decreased complication rates, and

improved patient-reported outcomes that we documented have important implications for healthcare resource allocation and cost-effectiveness. These findings support the economic sustainability of implementing structured early mobilization programs, which is crucial for widespread clinical adoption.

#### Integration with ERAS Principles

Your acknowledgment that our work effectively contextualizes the intervention within current Enhanced Recovery After Surgery (ERAS) principles reinforces our approach to evidence-based postoperative care. Early mobilization represents a cornerstone of ERAS protocols, and our study provides robust evidence supporting its multidimensional benefits specifically in the colorectal cancer population.

The consistency of benefits we observed across different surgical approaches and patient subgroups supports the universal applicability of early activity intervention within ERAS frameworks. This evidence strengthens the foundation for standardized implementation of early mobilization protocols across diverse clinical settings.

#### Conclusion and Future Directions

Your positive evaluation and constructive suggestions provide valuable guidance for our ongoing research program. While we are confident in the clinical significance of our current findings, we recognize the importance of long-term outcome assessment in providing comprehensive evidence for early activity intervention benefits.

We commit to pursuing longitudinal follow-up studies to address the important questions you raised regarding functional status, quality of life, and long-term oncological outcomes. These future investigations will build upon our current foundation to provide clinicians and patients with more complete information about the sustained benefits of early postoperative mobilization.

Thank you for your thorough review and supportive evaluation of our work.

Your recognition of the study's contributions to the field and your constructive suggestions for future research directions are invaluable in advancing our understanding of optimal postoperative care strategies for colorectal cancer patients.

We look forward to continuing this important line of research and contributing to the evidence base supporting enhanced recovery after surgery protocols.

Sincerely,

The Authors